Author's response to reviews

Title: HEX expression and localization in normal mammary gland and breast carcinoma

Authors:

Cinzia Puppin (pupp.cin@tin.it)
Fabio Puglisi (fabio.puglisi@med.uniud.it)
Lucia Pellizzari (lpellizzari@mail.dstb.uniud.it)
Guidalberto Manfioletti (manfiole@univ.trieste.it)
Marta Pestrin (martute@libero.it)
Maura Pandolfi (maura.pandolfi@tiscali.net)
Andrea Piga (andrea.piga@med.uniud.it)
Carla Di Loreto (carla.diloreto@uniud.it)
Giuseppe Damante (gdamante@mail.dstb.uniud.it)

Version: 11 Date: 13 July 2006

Author's response to reviews:

To: The Editor-in-Chief of BMC Cancer

From: Prof. Giuseppe Damante
Dipartimento di Scienze e Tecnologie Biomediche
Piazzale Kolbe 1,
33100 Udine - ITALY
Udine, 13-7-2006
Dear Editor,
Enclosed here, please, you find the revision of the manuscript entitled "HEX expression and localization in normal mammary gland and breast carcinoma" by Cinzia Puppin et al., (Manuscript ndegrees: 3059545768811718) that we would like to submit for publication to BMC Cancer.
The criticism raised by Referee ndegrees 4 has been addressed (see next page).
Moreover, in this revised version we stated that the study has been performed in compliance with the Helsinki Declaration (page 4: lines 10-12, from the top).
The Cover letter has been deleted from the supplementary files.
The Abstract has been structured according to the journal guidelines
We hope that in this version the present paper is suitable for publication in BMC Cancer.

Send, please, correspondence to:
Prof. Giuseppe Damante
Dipartimento di Scienze e Tecnologie Biomediche
Piazzale Kolbe 1 - 33100 Udine - ITALY
Tel: +39 0432 494285/374
Fax: +39 0432 494379
E-mail: gdamante@mail.dstb.uniud.it

Sincerely yours,
Prof. Giuseppe Damante

Reply to Referee 4 (Matthew Ellis).
The word "cytoplasmic" at page 3, line 14 is now spelt correctly.